Author's response to reviews

Title: Clinical Significance of Hypoalbuminemia in Outcome of Patients with Scrub Typhus

Authors:

Chang-Seop Lee (lcsmd@jbnu.ac.kr)
In-Suk Min (vargave8@hanmail.net)
Jeong-Hwan Hwang (smilehwang77@hanmail.net)
Keun-Sang Kwon (drkeunsang@jbnu.ac.kr)
Heung-Bum Lee (lhbmd@jbnu.ac.kr)

Version: 4 Date: 12 July 2010

Author's response to reviews:

Major revisions (we require the author to make these changes)

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution

-> We re-wrote

Competing interests: Please include a `Competing interests' section between the Conclusions and Authors' contributions. If there are none to declare, please write `The authors declare that they have no competing interests'.

Please see here for a full list of questions to consider when writing your competing interests statement-
http://www.biomedcentral.com/bmcmed/ifora/?txt_jou_id=2999&txt_mst_id=50733#interests

-> We wrote

Authors' contributions: Please include an Authors’ contributions section towards the end of the manuscript text, directly preceding the Acknowledgements.

Please use the following format as an example, using initials to refer to each author’s contribution: AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination.

-> We added.

Please state that all authors read and approved the final manuscript.